Author's response to reviews

Title: Looking inside the black box: A theory-based process evaluation alongside a randomised controlled trial of printed educational materials (the Ontario printed educational message, OPEM) to improve referral and prescribing practices in primary care in Ontario, Canada

Authors:

Jeremy M Grimshaw (jgrimshaw@ohri.ca)
Merrick Zwarenstein (merrick.zwarenstein@ices.on.ca)
Jacqueline M Tetroe (jitetroe@ohri.ca)
Gaston Godin (Gaston.Godin@fsi.ulaval.ca)
Ian D Graham (igraham@ohri.ca)
Louise Lemyre (louise.lemyre@uottawa.ca)
Martin P Eccles (martin.eccles@ncl.ac.uk)
Maire Johnston (m.johnston@abdn.ac.uk)
Jillian J Francis (j.francis@abdn.ac.uk)
Jan Hux (jan@ices.on.ca)
Keith O'Rourke (korourke@ohri.ca)
France Legare (France.Legare@mfa.ulaval.ca)
Justin Presseau (justin.presseau@alumni.uottawa.ca)

Version: 3 Date: 3 October 2007

Author's response to reviews:

The copy-edits have been accepted and the title and abstract have been updated in the online system accordingly.